Author's response to reviews

Title: Glucose Prediction by Analysis of Exhaled Metabolites - a Systematic Review

Authors:

Jan Hendrik Leopold (j.h.leopold@amc.uva.nl)
Roosmarijn van Hooijdonk (r.t.vanhooijdonk@amc.uva.nl)
Peter J Sterk (p.j.sterk@amc.uva.nl)
Ameen Abu-Hanna (a.abu-hanna@amc.uva.nl)
Marcus J Schultz (marcus.j.schultz@gmail.com)
Lieuwe DJ Bos (l.d.bos@amc.uva.nl)

Version: 2  Date: 5 March 2014

Author's response to reviews:

As requested by one of the reviewers, several sections are shortened and revised. First, one of the reviewers noted that Quadas-2-tool results found in table 2 were not clear. The table was probably altered by mistake after uploading the manuscript, but is adjusted to be more clear. Second, a part of results was transferred to the discussion section. Third, parts of the discussion were shortened and we referred to other literature that cover this information more thoroughly. Finally, the manuscript was checked for typographical and grammatical errors, and incomplete references were addressed.